Author’s response to reviews

Title: Tumor-directed gene therapy in mice using a composite nonviral gene delivery system consisting of the piggybac transposon and polyethylenimine

Authors:

Yu Kang (yukang@fudan.edu.cn)
Xiaoyan Zhang (xyzh03@163.com)
Wei Jiang (jw52317@hotmail.com)
Chaoqun Wu (cqwu@fudan.edu.cn)
Chunmei Chen (cmchen@fudan.edu.cn)
Yufang Zheng (zhengyf@fudan.edu.cn)
Jianren Gu (nlog@sh163.net)
Congjian Xu (xucj@hotmail.com)

Version: 4 Date: 20 March 2009

Author’s response to reviews:

We improved the grammar used in the manuscript.